Annals of Internal Medicine

Original Research

Weight and Metabolic Changes After Switching From Tenofovir Disoproxil Fumarate to Tenofovir Alafenamide in People Living With HIV

A Cohort Study

Bernard Surial, MD*; Catrina Mugglin, MD, MSc*; Alexandra Calmy, MD, PhD; Matthias Cavassini, MD; Huldrych F. Günthard, MD; Marcel Stöckle, MD; Enos Bernasconi, MD; Patrick Schmid, MD; Philip E. Tarr, MD; Hansjakob Furrer, MD; Bruno Ledergerber, PhD; Gilles Wandeler, MD, MSc†; and Andri Rauch, MD†, and the Swiss HIV Cohort Study‡

Background: Tenofovir-based antiretroviral therapy (ART) has become first-line in all major HIV treatment guidelines. Compared with tenofovir disoproxil fumarate (TDF), tenofovir alafenamide (TAF) has a favorable renal and bone safety profile, but concerns about metabolic complications remain.

Objective: To assess weight changes, the development of overweight/obesity, and changes in lipid levels 18 months after replacing TDF with TAF.

Design: Cohort study.

Setting: 5 university hospitals, affiliated hospitals, and private physicians in Switzerland.

Participants: 4375 adults living with HIV who received TDFcontaining ART for 6 months or longer.

Measurements: Changes in weight and lipid levels were assessed using mixed-effect models. Differences in proportions of newly overweight/obese participants were calculated using 2-proportions *Z* tests.

Results: 4375 individuals were included, with follow-up between 1 January 2016 and 31 July 2019. Median age was 50 years (interquartile range, 43 to 56 years), 25.9% were female, and 51.7% had a normal body mass index (BMI); 3484 (79.6%) switched to TAF and 891 (20.4%) continued TDF. After 18 months, switching to TAF was associated with an adjusted mean weight increase of 1.7 kg (95% CI, 1.5 to

enofovir plays an important role in antiretroviral therapy (ART) for people living with HIV (PLWH) and is recommended as part of the first-line regimens in all major HIV treatment guidelines (1-4). Tenofovir disoproxil fumarate (TDF) has been associated with proximal renal tubulopathy and loss of bone mineral density (5-7). The more favorable bone and renal safety profile of tenofovir alafenamide (TAF) (8-10) led to the replacement of TDF with TAF in most ART guidelines (2-4). However, TAF is not part of the World Health Organization's preferred first-line regimens due to concerns about metabolic side effects (1). In treatment-naive patients, TAF was associated with rising blood lipid levels and an increased need for lipid-lowering therapy compared with TDF (11), and recent data indicate that TAF leads to a substantially larger increase in weight compared with TDF in PLWH initiating ART (12, 13).

So far, weight and metabolic changes have been assessed mainly in ART-naive patients, which makes the

2.0 kg), compared with 0.7 kg (CI, 0.4 to 1.0 kg) with the continued use of TDF (between-group difference, 1.1 kg [CI, 0.7 to 1.4 kg]). Among individuals with a normal BMI, 13.8% who switched to TAF became overweight/obese, compared with 8.4% of those continuing TDF (difference, 5.4 percentage points [CI, 2.1 to 8.8 percentage points]). Switching to TAF led to increases in adjusted mean total cholesterol (0.25 mmol/L [9.5 mg/dL]), high-density lipoprotein cholesterol (0.05 mmol/L [1.9 mg/dL]), low-density lipoprotein cholesterol (0.12 mmol/L [4.7 mg/dL]), and triglyceride (0.18 mmol/L [16.1 mg/dL]) levels after 18 months.

Limitation: Short follow-up, small subgroup analyses, and potential residual confounding.

Conclusion: Replacing TDF with TAF is associated with adverse metabolic changes, including weight increase, development of obesity, and worsening serum lipid levels.

Primary Funding Source: Swiss National Science Foundation.

Ann Intern Med. doi:10.7326/M20-4853

Annals.org

For author, article, and disclosure information, see end of text. This article was published at Annals.org on 16 March 2021.

* Drs. Surial and Mugglin contributed equally to this work.

† Drs. Wandeler and Rauch contributed equally to this work.

‡ For members of the Swiss HIV Cohort Study, see the **Appendix** (available at Annals.org).

interpretation of results challenging. Because effective HIV treatment reduces the infection-associated catabolism, and thereby leads to weight increases in the first months after starting ART (14), it is difficult to differentiate between metabolic changes due to the return to health and adverse drug reactions in individuals initiating ART. In addition, most data on weight and metabolic changes were gathered among selected participants in clinical trials, and data from well-described real-world cohorts are scarce. We used data from the Swiss HIV Cohort Study (SHCS) to assess weight changes and metabolic outcomes in PLWH receiving stable ART who switched from TDF to TAF.

See also:

Summary for Patients

ORIGINAL RESEARCH

Methods

Study Population

The SHCS (www.shcs.ch) is an ongoing, nationally representative cohort study that was established in 1988. It includes close to 80% of all PLWH receiving ART in Switzerland who are followed in 1 of 5 university hospitals, 2 regional hospitals, or 15 affiliated hospitals or by 1 of 36 private physicians (15). Laboratory values, sociodemographic data, and clinical data are prospectively recorded at registration and every 6 months thereafter using a standardized protocol (http://shcs.ch/292-instructions). Assessments at every follow-up visit include weight measurement, documentation of all changes in medication (including ART and comedications), and glucose and lipid measurements at accredited laboratories (https://shcs.ch/173-laboratories). Data quality and consistency are ensured by quality checks and regular site visits of participating centers. All centers' local ethical committees approved the cohort study, and all patients provided written informed consent.

For the present study, we considered all participants with follow-up between January 2016, the year in which TAF was approved in Switzerland, and 31 July 2019 (database closure). We included patients who were receiving a TDF-containing treatment for at least 6 months and either continued TDF until the end of the study period or had TDF replaced by TAF. We defined the index visit as the switching date for patients who had TDF replaced with TAF, and a random sample of these switching dates was selected and assigned to individuals who continued TDF. All individuals with any follow-up after the index visit were included, and follow-up of individuals who stopped TAF before the end of the observation period was censored at that time. We excluded patients who received different nucleoside reverse transcriptase inhibitors (NRTIs) between the use of TDF and TAF and women who were pregnant during the study period.

Outcomes and Definitions

The primary aims were to compare weight trajectories over time between individuals who continued TDF and those who had TDF replaced by TAF and to estimate the differences in weight between the index visit and 18 months thereafter. To account for different changes in weight before the index visit, we included all weight measurements from 2.5 years before that date until the end of each individual's follow-up. The main exposure of interest was switching from TDF to TAF. Secondary outcomes were the proportion of individuals who became overweight or obese (body mass index [BMI] >25 kg/m²), mean changes in lipid levels (total cholesterol, low-density lipoprotein [LDL] cholesterol, highdensity lipoprotein [HDL] cholesterol, triglycerides, and total cholesterol-HDL ratio), and the incidence of diabetes after the index visit. Weight categories were classified according to BMI as normal (18.5 to 24.9 kg/m²), overweight (25 to 29.9 kg/m²), obese (\geq 30 kg/m²), and underweight (<18.5 kg/m²) (16). Diabetes at the index visit was defined as a hemoglobin A_{1c} level of 6.5% or greater or treatment with antidiabetic medication, and history of cardiovascular disease included myocardial

infarction, cerebral infarction, coronary angioplasty or stenting, coronary artery bypass grafting, or any procedure on peripheral arteries. Incident diabetes was defined as a hemoglobin A_{1c} level of 6.5% or greater, a fasting glucose level greater than 7.0 mmol/L, or random glucose levels greater than 11.0 mmol/L on at least 2 visits or the start of use of any antidiabetic drug after the index visit. The occurrence of cardiovascular events or elevations of hemoglobin A_{1c} are reported by the treating cohort physician using standardized forms.

Statistical Analysis

Patient characteristics were compared between individuals who continued TDF and those who switched to TAF using χ^2 and Wilcoxon rank-sum tests. Crude weight trajectories over time were presented using locally estimated scatter plot smoothing. Adjusted mean changes in weight over time in absolute values were estimated using 2-level multivariable mixed-effect models, with random intercepts for each individual nested within the corresponding cohort site and random slopes for time on the individual level. To allow the weight trajectories to be nonlinear, time was modeled using restricted cubic splines with the numbers of knots chosen to minimize Bayesian information criteria. The first knot was positioned at the index visit because trajectories were expected to change at that time, and the remainder of the knots were evenly spaced at percentiles 33.3 and 66.6 of the available follow-up time thereafter (0, 7.1, and 14.8 months). Because all individuals were receiving TDF before the index visit, and crude weight trajectories before the index visit were similar between individuals who continued TDF and those who switched to TAF after the index visit, preindex weight trajectories for both groups were combined to improve model fit. Covariates were prespecified characteristics that were potentially associated with weight increase, the decision to switch to TAF, or both. Multivariable analyses for weight were adjusted for sex, African origin, age, time since ART initiation, CD4 cell count, and index visit values of weight and estimated glomerular filtration rate (calculated using the Chronic Kidney Disease Epidemiology Collaboration formula). In addition, we included the third ART drug component used after the switch, categorized as integrase strand transfer inhibitors (INSTIs), nonnucleoside reverse transcriptase inhibitors (NNRTIs), or protease inhibitors (PIs). The self-reported use of weight-modifying drugs (for example, antidiabetics, neuroleptics, systemic corticosteroids [17]), smoking status (yes or no), and physical activity (exercising more than twice a week, 1 to 2 times per week, 1 to 4 times per month, or never) were included as time-varying covariates. We assessed model fit with Bayesian information criteria and compared models using likelihood ratio tests. Because there was evidence for effect modification by weight at the index visit, we included an interaction term with this variable and showed trajectories stratified by 3 weight categories based on index weight tertiles. In exploratory analyses, the joint effect of sex and African origin and the influence

of the third drug used in the ART regimen after the index visit on the association between the main exposure variable and weight was assessed using interaction terms. Interactions with sex, African origin, and the third drug did not improve model fit, so these variables were included only as covariates in the final model. Differences in proportions of individuals who became overweight or obese were calculated using 2-proportions *Z* tests.

Mean differences in serum lipid levels were estimated similarly to weight analyses using the same random-effects structure and were adjusted for age, sex, African origin, and individual lipid level at the index visit and for the time-varying values of weight, physical activity, and the use of lipid-lowering drugs (including statins, fibrates, and nicotinic acid). Analyses on incidence rate ratios of new-onset diabetes were restricted to individuals without diabetes at the index visit and were calculated using quasi-Poisson regression adjusted for age, sex, African origin, and BMI, including an offset for the total number of follow-up years. All statistical analyses were performed using R, version 4.0.0 (18).

Sensitivity Analyses

Weight analyses restricted to individuals with at least 6 and 12 months of follow-up were performed to examine potential attrition bias. To limit the effect of replicating HIV infection on weight, we conducted a sensitivity analysis restricted to individuals with an HIV viral load of less than 50 copies/mL during the prior year and throughout follow-up. To minimize confounding by other ART components, we performed an analysis restricted to individuals who had TDF replaced with TAF, with the rest of the regimen left unchanged. We repeated the weight analysis in those who switched from an abacavir (ABC)containing ART to TAF to investigate whether weight changes are associated with starting TAF or stopping TDF, as a protective effect of the latter on serum lipid levels has been shown previously (19).

Role of the Funding Source

The present study was funded by the SHCS, supported by the Swiss National Science Foundation. The funders had no role in the study design, data collection, analysis, interpretation, or writing of the manuscript. The corresponding author had full access to all of the data and was responsible for the decision to submit this article.

RESULTS

Study Population

Of 10674 participants under follow-up, 8047 had received TDF for more than 6 months. We excluded 3149 individuals who switched from TDF to a different NRTI, 474 without follow-up after the index visit, and 49 women who became pregnant. The study population included 4375 individuals; the median age was 50 years (interquartile range [IQR], 43 to 56 years), 25.9% were female, and 51.7% had a normal BMI. Of the 4375 participants, 891 (20.4%) continued TDF until the end of the study and 3484 (79.6%) switched to TAF (Appendix **Figure 1**, available at Annals.org). Patients who switched to TAF were more likely to be male and men having sex with men, were less likely to be of African origin, had a lower estimated glomerular filtration rate, and were more likely to receive INSTI-based regimens than those who continued TDF (Table 1). The median follow-up was 17.1 months (IQR, 9.6 to 21.3 months) in individuals who switched to TAF and 17.5 months (IQR, 13.0 to 21.2 months) in those who continued TDF. Individuals in both groups were assessed at a median of 3 follow-up visits after the index visit, and 20 (2.2%) who continued TDF and 60 (1.7%) who switched to TAF were lost to follow-up. Observations were missing in less than 2% of visits for all covariates (Appendix Table 1, available at Annals.org).

Changes in Weight

Crude weight trajectories were similar before the index visit and diverged thereafter (Figure 1, top; Appendix Figure 2, available at Annals.org). In unadjusted analyses, switching to TAF was associated with a mean weight increase of 1.8 kg (95% Cl, 1.6 to 2.0 kg) 18 months after the index visit, compared with 0.7 kg (Cl, 0.4 to 1.0 kg) with the continuous use of TDF (betweengroup difference, 1.1 kg [Cl, 0.7 to 1.4 kg]). These estimates remained similar after adjustment for confounders (Table 2; Figure 1, middle; Appendix Table 2 and Appendix Figure 3, available at Annals.org). Figure 1 (bottom) describes changes in BMI categories at 18 months compared with the index visit in all patients with available weight measurements. Among individuals with a normal BMI at the index visit and available follow-up at 18 months, 211 of 1529 (13.8%) who switched to TAF became overweight or obese after 18 months, compared with 35 of 419 (8.4%) who continued TDF (difference, 5.4 percentage points [Cl, 2.1 to 8.8 percentage points]).

The use of TAF was associated with statistically significant increases in adjusted mean weight regardless of sex or origin (Table 2). Between-group weight differences of TAF compared with TDF were most pronounced among women of African origin (1.5 kg [CI, 0.4 to 2.5 kg]), followed by women of non-African origin (1.4 kg [CI, 0.2 to 2.7 kg]) and men of non-African origin (1.1 kg [CI, 0.7 to 1.5 kg]), and were not significant among men of African origin (P < 0.001 for the joint interaction with African origin and sex). Weight increases while receiving TAF were observed regardless of the third drug used after the index visit (P = 0.055 for interaction [Table 2]), and the magnitude of weight increases diminished with higher index visit weight and BMI (Figure 2; Table 2; Appendix Figure 4, available at Annals.org).

Sensitivity Analyses of Weight Changes

Analyses based on individuals with at least 6 and 12 months of follow-up showed similar weight changes as the main analysis (Appendix Table 3, available at Annals .org). In an analysis restricted to individuals with continuously suppressed HIV viral load, switching to TAF remained associated with an adjusted weight increase of 1.8 kg (Cl, 1.6 to 2.1 kg), compared with 0.5 kg (Cl, 0.2 to 0.8 kg) in those continuing TDF (Appendix Figure 5, available at

Table 1. Characteristics of the Study Population at the Index Visit

$\begin{split} & \text{Wereen, } n(\%) & 272 (30.5) & 861 (24.7) & <0.001 \\ & \text{Median age (IQR), } & 49 (42.55) & 50 (43.56) & 0.017 \\ & \text{African origin, } n(\%) & 180 (20.2) & 462 (13.3) & <0.001 \\ & \text{Tarsmission group, } n(\%) & & & & <0.001 \\ & \text{Tarsmission group, } n(\%) & 1732 (49.7) & & <0.001 \\ & \text{Men having sex with men} & 373 (41.9) & 1732 (49.7) & & <0.001 \\ & \text{Men having sex with men} & 373 (41.9) & 1732 (49.7) & & & <0.001 \\ & \text{Men having sex with men} & 373 (41.9) & 1732 (49.7) & & & & <0.001 \\ & \text{Median CD4 count (IQR), cells/µL* & 637 (485-824) & 658 (498-859) & 0.097 \\ & \text{Median CD4 count (IQR), cells/µL* & 637 (485-824) & 658 (498-859) & 0.097 \\ & \text{Median duration of TDF theore index visit (IQR), y & 11.3 (65-18.3) & 11.4 (66-19.3) & 0.54 \\ & \text{Median duration of TDF theore index visit (IQR), y & 77 (5.1-11.0) & 8.2 (52-10.0) & 0.21 \\ & \text{Median duration of TDF theore index visit (IQR), y & 77 (5.5-11.0) & 8.2 (52-10.0) & 0.21 \\ & \text{Median duration of TDF theore index visit (IQR), y & 77 (5.5-11.0) & 8.2 (52-10.0) & 0.23 \\ & \text{Median duration of TDF theore index visit (IQR), y & 73 (55-55 & 240 (6.9) & 0.157 \\ & \text{Smoker, } n(\%) & 31 (37.1) & 130 (37.6) & 0.138 \\ & Receiving lipid-lowering therapy, n(\%) & 31 (37.1) & 130 (37.6) & 0.138 \\ & \text{Receiving lipid-lowering therapy, n(\%) & 31 (37.1) & 130 (37.6) & 0.55 \\ & \text{Median weight (IQR), kg & 74 (65-84) & 74 (65-84) & 0.52 \\ & \text{Median Weight (IQR), kg & 74 (55-84) & 74 (55-84) & 0.52 \\ & \text{Median weight (IQR), kg & 74 (55-84) & 74 (55-84) & 0.52 \\ & \text{Median weight (IQR), kg & 73 (45 (24) & 132 (38.3) & 0.44 \\ & \text{Outerweight (25-29 kg/m^2) & 272 (30.5) & 1097 (31.5) & 0.45 \\ & \text{Merian total cholesterol level (WR) & & & & & & & & & & & & & & & & & & &$	Characteristic	Continued TDF (n = 891)	Switched to TAF (n = 3484)	P Value
Median age (IQR), y 49 (42-55) 50 (43-56) 0.017 African origin, n (%) 180 (20.2) 462 (13.3) <0.001	Women, <i>n (%)</i>	272 (30.5)	861 (24.7)	< 0.001
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Median age (IQR), y	49 (42-55)	50 (43-56)	0.017
Transmission group, n (%) <.0.001	African origin, n (%)	180 (20.2)	462 (13.3)	< 0.001
Men having sex with men 373 (41.9) 1732 (49.7) People who inject drugs 77 (8.6) 356 (10.2) Other 441 (49.5) 1396 (40.1) Median CD4 count (ICR), cells/µL* 637 (485-824) 658 (498-859) 0.097 Median duration of ART before index visit (ICR), y 11.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of ART before index visit (ICR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median eGFR, (ICR), mL/min 96.0 (82.0-107.0) 87.9 (74.8-101.1) <0.001	Transmission group, n (%)			< 0.001
People who inject drugs 77 (8.6) 356 (10.2) Other 441 (49.5) 1396 (40.1) Median CD4 count (IQR), cells/µL* 637 (485-824) 658 (498-859) 0.097 Median CD4 adir (IQR), cells/µL* 221 (107-323) 215 (113-325) 0.98 Median duration of ART before index visit (IQR), y 17.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median duration of TDF therapy before index visit (IQR), y 7.4 (5.5-18.3) 11.4 (6.6-19.3) 0.23 Median BUTCY of CV disease, n (%) 38 (4.3) 186 (5.3) 0.23 Mistory of CV disease, n (%) 331 (37.1) 1309 (37.6) 0.138 Receining lipid-lowering therapy, n (%) 130 (14.6) 555 (15.9) 0.35 Median weight (IQR), kg 74 (65-84) 74 (65-84) 0.52 Median Weight (1QR), kg/m ²) 272 (30.5) 1097 (31.5) 0.44 Overweight (25-29.9 kg/m ²) 272 (30.5)	Men having sex with men	373 (41.9)	1732 (49.7)	
Other 441 (49.5) 1396 (40.1) Median CD4 count (IQR), cells/µL* 637 (485-824) 658 (498-859) 0.097 Median duration of ART before index visit (IQR), y 11.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of DTP therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median eGFR, (IQR), mL/min 96.0 (82.0-107.0) 87.9 (74.8-101.1) <0.001	People who inject drugs	77 (8.6)	356 (10.2)	
Median CD4 count (IQR), cells/µL* 637 (485-824) 658 (498-859) 0.097 Median CD4 nadir (IQR), cells/µL* 221 (109-323) 215 (113-325) 0.98 Median duration of ART before index visit (IQR), y 11.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.23 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.23 Median weffer, (IQR), m/min 960 (82.0-107.0) 8.7 (74.8-10.1) <0.001	Other	441 (49.5)	1396 (40.1)	
Median CD4 nadir (IQR), cells/µL* 221 (109-323) 215 (113-325) 0.98 Median duration of ART before index visit (IQR), y 11.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median eGFR, (IQR), mL/min 96.0 (82.0-107.0) 87.9 (74.8-101.1) <0.001	Median CD4 count (IQR), <i>cells/µL</i> *	637 (485-824)	658 (498-859)	0.097
Median duration of ART before index visit (IQR), y 11.3 (6.5-18.3) 11.4 (6.6-19.3) 0.54 Median duration of DF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Median duration of TDF therapy before index visit (IQR), y 7.7 (5.1-11.0) 8.2 (5.2-11.0) 0.21 Diabetes, n (%) 38 (4.3) 186 (5.3) 0.23 History of CV disease, n (%) 38 (4.3) 186 (5.3) 0.23 Smoker, n (%) 33 (37.1) 1309 (37.6) 0.138 Receiving lipid-lowering therapy, n (%) 130 (14.6) 555 (15.9) 0.35 Median Weight (IQR), kg/m ² 24.5 (22.1-27.5) 24.4 (22.1-27.3) 0.49 BMI category at the index visit, n (%)	Median CD4 nadir (IQR), cells/µL*	221 (109-323)	215 (113-325)	0.98
Median duration of TDF therapy before index visit (IQR), y 7, 7 (5, 1-11.0) 8.2 (5.2-11.0) 0.21 Median eGFR, (IQR), mL/min 96.0 (82.0-107.0) 87.9 (74.8-101.1) <0.001	Median duration of ART before index visit (IQR), y	11.3 (6.5-18.3)	11.4 (6.6-19.3)	0.54
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Median duration of TDF therapy before index visit (IQR), y	7.7 (5.1-11.0)	8.2 (5.2-11.0)	0.21
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Median eGFR, (IQR), <i>mL/min</i>	96.0 (82.0-107.0)	87.9 (74.8-101.1)	< 0.001
History of CV disease, n (%) 49 (5.5) 240 (6.9) 0.157 Smoker, n (%) 331 (37.1) 1309 (37.6) 0.138 Receiving lipid-lowering therapy, n (%) 130 (14.6) 555 (15.9) 0.35 Median weight (IQR), kg 74 (65-84) 74 (65-84) 0.52 Median BMI (IQR), kg/m^2 24.5 (22.1-27.5) 24.4 (22.1-27.3) 0.49 BMI category at the index visit, n (%)	Diabetes, n (%)	38 (4.3)	186 (5.3)	0.23
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	History of CV disease, n (%)	49 (5.5)	240 (6.9)	0.157
Receiving lipid-lowering therapy, n (%) 130 (14.6) 555 (15.9) 0.35 Median weight (IQR), kg 74 (65-84) 74 (65-84) 0.52 Median BMI (IQR), kg/m^2 22.1 (22.1 - 27.5) 24.4 (22.1 - 27.3) 0.49 BMI category at the index visit, n (%)	Smoker, n (%)	331 (37.1)	1309 (37.6)	0.138
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Receiving lipid-lowering therapy, n (%)	130 (14.6)	555 (15.9)	0.35
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Median weight (IQR), kg	74 (65-84)	74 (65-84)	0.52
BMI category at the index visit, n (%) 0.147 Normal (18.5-24.9 kg/m ²) 445 (49.9) 1816 (52.1) Overweight (25-29.9 kg/m ²) 272 (30.5) 1097 (31.5) Obese (≥30 kg/m ²) 122 (13.7) 405 (11.6) Underweight (18.5 kg/m ²) 37 (4.2) 132 (3.8) Third drug used at index visit, n (%) 74 (2.2) 1333 (38.3) PI 84 (9.4) 741 (21.3) NNRTI 606 (68.0) 1390 (39.9) Median total cholesterol level (IQR) 0.73 mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (12-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 108 (87-132) 106 (83-130) Median HDL cholesterol level (IQR) 0.136 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (87-132) 106 (83-130) <	Median BMI (IQR), kg/m^2	24.5 (22.1-27.5)	24.4 (22.1-27.3)	0.49
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	BMI category at the index visit, n (%)	, ,	· · ·	0.147
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Normal (18.5–24.9 kg/m ²)	445 (49.9)	1816 (52.1)	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Overweight (25-29.9 kg/m ²)	272 (30.5)	1097 (31.5)	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Obese (≥30 ka/m²)	122 (13.7)	405 (11.6)	
Third drug used at index visit, n (%)	Underweight (<18.5 kg/m ²)	37 (4.2)	132 (3.8)	
INSTI 189 (21.2) 1333 (38.3) PI 84 (9.4) 741 (21.3) NNRTI 606 (68.0) 1390 (39.9) Median total cholesterol level (IQR) 0.73 mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4 (6) 3.8 (3.1-4.7) 0.039	Third drug used at index visit. n (%)		- ()	< 0.001
PI 84 (9.4) 741 (21.3) NNRTI 606 (68.0) 1390 (39.9) Median total cholesterol level (IQR) 0.73 mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mg/dL 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4 (6) 3.8 (3.1-4.7) 0.039	INSTI	189 (21.2)	1333 (38.3)	
NNRTI 606 (68.0) 1390 (39.9) Median total cholesterol level (IQR) 0.73 mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4 (6) 3.8 (3.1-4.7) 0.039	PI	84 (9,4)	741 (21.3)	
Median total cholesterol level (IQR) 0.73 mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 9.27 (2.2-3.4) 0.136 mmol/L 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mg/dL 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4 (6) 3.8 (3.1-4.7) 0.039	NNRTI	606 (68.0)	1390 (39.9)	
mmol/L 4.9 (4.2-5.5) 4.8 (4.2-5.5) mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4 (6) 3.8 (3.1-4.7) 0.039	Median total cholesterol level (IOR)		,	0.73
Instruction Instruction Instruction mg/dL 186 (16-214) 186 (162-214) Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	mmol/L	4.9 (4.2-5.5)	4.8 (4.2-5.5)	
Median HDL cholesterol level (IQR) 0.070 mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	ma/dL	186 (16-214)	186 (162-214)	
mmol/L 1.3 (1.1-1.6) 1.3 (1.0-1.6) mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	Median HDL cholesterol level (IOR)			0.070
mg/dL 50 (41-61) 49 (39-60) Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	mmol/L	1.3 (1.1-1.6)	1.3 (1.0-1.6)	
Median LDL cholesterol level (IQR) 0.136 mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	ma/dL	50 (41-61)	49 (39-60)	
mmol/L 2.8 (2.2-3.4) 2.7 (2.2-3.4) mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	Median LDL cholesterol level (IOR)			0.136
mg/dL 108 (87-132) 106 (83-130) Median triglyceride level (IQR)	mmol/L	2.8 (2.2-3.4)	2.7 (2.2-3.4)	
Median triglyceride level (IQR) mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IQR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	ma/dl	108 (87-132)	106 (83-130)	
mmol/L 1.2 (0.9-1.9) 1.4 (1.0-2.1) mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IOR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	Median triglyceride level (IOR)			< 0.001
mg/dL 108 (80-168) 124 (89-186) Median total cholesterol-HDL ratio (IOR) 3.7 (3.0-4.6) 3.8 (3.1-4.7) 0.039	mmol/L	1.2 (0.9-1.9)	1.4 (1.0-2.1)	
Median total cholesterol-HDL ratio (IOR) 37 (30-4 6) 38 (3 1-4 7) 0.039	ma/dL	108 (80–168)	124 (89-186)	
	Median total cholesterol-HDL ratio (IQR)	3.7 (3.0-4.6)	3.8 (3.1-4.7)	0.039

ART = antiretroviral therapy; BMI = body mass index; CV = cardiovascular; eGFR = estimated glomerular filtration rate; HDL = high-density lipoprotein; INSTI = integrase strand transfer inhibitor; IQR = interquartile range; LDL = low-density lipoprotein; NNRTI = nonnucleoside reverse transcriptase inhibitor; PI = protease inhibitor; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate. * To convert to SI units (× 10⁹ cells/L), multiply by 0.001.

Annals.org). When including only individuals who had TAF replaced by TDF without further changes in ART, TAF remained associated with significant increases in weight, regardless of the third drug used (**Appendix Table 4**, available at Annals.org). Finally, switching from ABC to TAF was associated with a weight increase of 1.7 kg (Cl, 1.0 to 2.4 kg) after 18 months, compared with 0.5 kg (Cl, 0.3 to 0.7 kg) with the continuous use of ABC (between-group difference, 1.2 kg [Cl, 0.5 to 1.9 kg]) (Figure 3).

Lipid and Glucose Levels

At the index visit, triglyceride levels and total holesterol-HDL ratios were slightly higher in individuals who switched to TAF than in those who continued TDF, whereas total cholesterol, HDL cholesterol, and LDL cholesterol levels were similar between groups (Table 1). Observations were missing in less than 2.5% of all follow-up visits (Appendix Table 1). Eighteen months after the index visit, switching to TAF was associated with

increases in total cholesterol, HDL cholesterol, LDL cholesterol, and triglyceride levels, whereas decreases in total cholesterol and LDL cholesterol levels were observed with the continuous use of TDF (Figure 4; Appendix Table 5, available at Annals.org). During followup, 127 individuals (3.6%) who switched to TAF started a new lipid-lowering treatment, compared with 30 (3.4%) who continued using TDF (difference, -0.3 percentage point [Cl, -1.7 to 1.1 percentage points]).

Among 4151 individuals without diabetes at the index visit, 4150 (99.9%) contributed to 5616 personyears of follow-up. The crude incidence rate of newonset diabetes among individuals who switched to TAF was 1.1 per 100 person-years compared with 0.9 per 100 person-years among those who continued TDF (unadjusted incidence rate ratio, 1.2 [CI, 0.6 to 2.6]). After adjustment for age, sex, African origin, and BMI at the index visit, the incidence rate ratio for new-onset diabetes was 1.3 (CI, 0.7 to 2.8) (Appendix Table 6, available

Figure 1. Changes in weight and BMI categories over time.



BMI = body mass index; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate. **Top.** Crude mean weight trajectories (*line*) and 95% CIs (*shaded area*) before and after the index visit for individuals who continued TDF and those who switched to TAF, derived using locally estimated scatter plot smoothing. **Middle.** Mean changes in weight (*line*) and 95% CIs (*shaded area*) compared with the index visit after switching to TAF versus continuing TDF. Adjusted for sex; African origin; age; CD4 cell count; time since antiretroviral therapy initiation; index visit values of weight and estimated glomerular filtration rate; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. Because all individuals were receiving TDF before the index visit, and preindex trajectories of individuals who switched to TAF and those who continued TDF were similar, their preindex visit trajectories were combined in the multivariable model. A total of 4291 individuals were included in the models. Knots were positioned at 0, 7.1, and 14.8 mo. **Bottom**. Unadjusted changes in the proportion of BMI categories after 18 mo compared with the index visit, stratified by whether individuals switched to TAF or continued TDF. Analysis restricted to individuals with available weight measurements at 18 mo (n = 2484).

>80 kg

0.8 (0.1 to 1.4)

0.015

Variable	Change in Weight After 18 mo of TDF (95% CI), <i>kg</i>	Change in Weight After 18 mo of TAF (95% CI), <i>kg</i>	Difference Between TAF and TDF (95% CI), <i>kg</i>	<i>P</i> Value for Difference
Overall sample	0.7 (0.4 to 1.0)	1.7 (1.5 to 2.0)	1.1 (0.7 to 1.4)	< 0.001
Sex and origin				
Women of non-African origin	-0.1 (-1.1 to 0.9)	1.4 (0.6 to 2.2)	1.4 (0.2 to 2.7)	0.026
Men of non-African origin	0.6 (0.2 to 0.9)	1.7 (1.4 to 2.0)	1.1 (0.7 to 1.5)	<0.001
Women of African origin	1.0 (0.2 to 1.8)	2.4 (1.7 to 3.2)	1.5 (0.4 to 2.5)	0.009
Men of African origin	0.9 (-0.2 to 2.1)	1.6 (0.5 to 2.6)	0.6 (-1.0 to 2.2)	0.44
Third drug used				
INST	0.1 (-0.5 to 0.7)	1.8 (1.5 to 2.1)	1./ (1.0 to 2.4)	< 0.001
NNRTI	0.8 (0.5 to 1.2)	1.9 (1.4 to 2.4)	1.1 (0.4 to 1.7)	0.001
PI	0.8 (-0.2 to 1.7)	1.8 (1.2 to 2.4)	1.1 (0.0 to 2.1)	0.061
Weight at index visit				
<70 kg	1.1 (0.6 to 1.6)	2.2 (1.8 to 2.6)	1.1 (0.5 to 1.7)	< 0.001
70-80 kg	0.3 (-0.2 to 0.8)	1.6 (1.2 to 2.0)	1.3 (0.7 to 1.9)	< 0.001

Table 2. Adjusted Changes in Weight From the Index Visit to 18 Months Thereafter in the Overall Sample and Across Subgroups*

INSTI = integrase strand-transfer inhibitor; NNRTI = nonnucleoside reverse transcriptase inhibitor; PI = protease inhibitor; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

1.3 (0.9 to 1.7)

* Adjusted for age, sex, African origin, body mass index, and CD4 cell count at the index visit; time since antiretroviral therapy initiation; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. Models include random intercepts for each patient nested within cohort site and a random slope for time on the individual level.

Figure 2. Changes in weight over time, stratified by the weight at the index visit.

0.5 (0.1 to 1.0)



Mean changes in weight (*line*) and corresponding 95% Cls (*shaded area*) after switching to TAF compared with continuing TDF, stratified by the weight at the index visit. Results are based on a multivariable model adjusted for sex; African origin; age; CD4 cell count; time since antiretroviral therapy initiation; index visit values of weight and estimated glomerular filtration rate; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. Because all individuals were receiving TDF before the index visit, and preindex trajectories of individuals who switched to TAF and those who continued TDF were similar, their preindex visit trajectories were combined in the multivariable model. A total of 4291 individuals were included in the model. Knots were positioned at 0, 7.1, and 14.8 mo. TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

6 Annals of Internal Medicine

at Annals.org). There was no evidence that switching from TDF to TAF increased incidence of diabetes among individuals with a higher BMI at the index visit (*P* for interaction = 0.95) (Appendix Figure 6, available at Annals. org).

DISCUSSION

In this nationwide cohort study, individuals switching from TDF to TAF experienced a larger weight increase than those who continued TDF over 18 months of followup. The largest difference between groups was observed among women of African (1.5 kg) and non-African (1.4 kg) origin. Compared with individuals who continued TDF, those who switched to TAF were more likely to become overweight and to experience worsening of serum lipid levels. Our estimates were robust across subgroups of patients regardless of whether TAF was administered together with PIs, NNRTIs, or INSTIs. Taken together, our results highlight the need for continuous monitoring of metabolic comorbid conditions during TAF-containing ART and for further exploring the mechanisms leading to metabolic changes in this population.

Compared with previous publications (identified by an English-language MEDLINE search to "[TAF or tenofovir alafenamid*] AND HIV AND weight"), the weight increase observed in our study is in line with 2 retrospective studies of patients switching from TDF to TAF (20, 21). However, compared with our study, the sample sizes were small (241 and 305, respectively), and the analyses were not adjusted for important confounders. In our study, switching to TAF compared with continuing TDF translated into a larger proportion of individuals with a normal BMI at the index visit who became overweight or obese during the study period, which might further contribute to increasing obesity rates among PLWH (22). The absolute increase in weight on TAF was largest among women of African origin, a finding that was also shown in a large pooled analysis of 8 clinical trials including 5680 treatment-naive PLWH (13). Whereas the underlying mechanisms remain unclear, women of African origin were also at higher risk for obesity among PLWH in a study from the United Kingdom (23). Although less marked than among women of African origin, weight increases among other demographic groups receiving TAF were statistically significant, generally exceeding 1.5 kg after 18 months.

In our study, weight increase while receiving TAF was observed with the concurrent use of all major third drug classes (PIs, NNRTIs, and INSTIs). Whereas a study in treatment-experienced patients showed similar weight changes regardless of the third drug class (24), studies from treatment-naive patients showed larger weight increases among patients receiving INSTI-based regimens (12, 13). Although an additional influence of other drugs, such as INSTI, cannot be excluded, the consistency of our findings across treatment regimens speaks for the important role of switching from TDF to TAF in driving weight increases. Increases in weight while receiving TAF without other ART components have also been observed in a study evaluating TAF-emtricitabine

for HIV preexposure prophylaxis (25). Finally, our observation that switching from ABC to TAF was also associated with weight increases further suggests that the increases seen after the replacement of TDF by TAF cannot only be attributed to stopping TDF.

We observed increased lipid levels among individuals who switched to TAF compared with those who continued TDF. These findings confirm and extend observations from registration trials and cohort studies, which consistently showed worsening lipid profiles (9, 26, 27) and an increased demand for lipid-lowering therapy with TAF (11). Several studies indicate that the increase in lipid levels in individuals switching from TDF to TAF might be attributed to stopping TDF, which has an intrinsic lipidlowering effect (19). Although weight increase and dyslipidemia can affect insulin resistance, we found no clear evidence for increased rates of new-onset diabetes with the use of TAF during the study. Neither TDF nor TAF itself led to insulin resistance among healthy volunteers (28, 29). However, follow-up data from the ADVANCE trial indicate that the large increases in weight observed with the use of TAF led to increased rates of diabetes (30).

Our study is among the largest to date investigating the effect of switching from TDF to TAF on weight and metabolic outcomes within a well-defined and nationally

Figure 3. Changes in weight over time, analysis of patients who received ABC and continued ABC (n = 2560) or switched to TAF (n = 427).



Mean changes in weight (*line*) and corresponding 95% CIs (*shaded area*) after switching to TAF compared with continuing ABC, adjusted for sex; African origin; age; CD4 cell count; time since antiretroviral therapy initiation; index visit values of weight and estimated glomerular filtration rate; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. Because all individuals were receiving ABC before the index visit, and preindex trajectories of individuals who switched to TAF and those who continued ABC were similar, their preindex visit trajectories were combined in the multivariable model. Knots were positioned at 0, 7.1, and 14.8 mo. ABC = abacavir; TAF = tenofovir alafenamide.





Mean changes (squares and diamonds) and 95% CIs (vertical line) in blood lipid values from the index visit to 18 mo thereafter, adjusted for age; sex; African origin; individual lipid levels at the index visit; and time-varying physical activity, weight, and use of lipid-lowering drugs. The models include random intercepts for each patient nested within cohort site and a random slope for time on the individual level. A total of 4290 individuals were included in the models. HDL = high-density lipoprotein; LDL = low-density lipoprotein; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

representative cohort. Assessing these outcomes among treatment-experienced PLWH allowed us to avoid the influence of the return to health, which inherently complicates the interpretation of studies among treatmentnaive individuals. We adjusted our analyses for a wide range of confounders, including time-updated physical activity and comedications, and our findings remained robust across several sensitivity analyses, including an analysis among patients who switched from ABC to TAF.

Some limitations of our study should be noted. Follow-up was relatively short to grasp the full effect of TAF on new-onset diabetes and lipid metabolism. In addition, subgroup analyses in specific demographic groups were based on small numbers, which limited our ability to detect differences. Information on physical activity and use of weight-modifying drugs was selfreported, and misclassification of these covariates may have affected our results. Although most individuals had replaced TDF with TAF at a time when no published evidence for a treatment-associated weight increase was available, confounding by indication cannot be fully excluded. Assuming that individuals who were more prone to weight increase continued TDF to avoid metabolic side effects of TAF, the difference in weight increase between TDF and TAF would have been underestimated. Third drug classes at the index visit differed markedly between groups, and our methods might have been insufficient to fully adjust for these imbalances. However, a sensitivity analysis restricted to individuals who only replaced TDF with TAF without further ART

modifications showed results consistent with our primary analysis. Finally, unmeasured residual confounding cannot be excluded in this observational study. Accordingly, weight increases after switching from TDF to TAF should also be explored in large-scale randomized controlled trials with sufficient follow-up.

In conclusion, our results indicate that switching from TDF to TAF is associated with metabolic adverse events, including obesity and dyslipidemia. Recommendations on the use of TAF should balance its advantages (renal and bone safety) with its potential harms, including metabolic complications. The decision to prefer TAF over TDF as a component of ART should be individualized and accompanied by the repeated assessment of cardiometabolic risk factors, including weight and lipids. Further studies are needed to provide more insight into the mechanisms of weight increase and metabolic effects of modern HIV drugs, to identify individuals at highest risk for such metabolic complications, and to assess the effect of these metabolic complications on clinical outcomes.

From Inselspital, Bern University Hospital, University of Bern, Bern, Switzerland (B.S., C.M., H.F., G.W., A.R.); Geneva University Hospital, University of Geneva, Geneva, Switzerland (A.C.); University Hospital of Lausanne, University of Lausanne, Lausanne, Switzerland (M.C.); University Hospital Zurich, University of Zurich, Zurich, Switzerland (H.F.G., B.L.); University Hospital Basel, University of Basel, Basel, Switzerland (M.S.); Regional Hospital of Lugano, University of Geneva, and University of Southern Switzerland, Lugano, Switzerland (E.B.); Cantonal Hospital of St. Gallen, St. Gallen, Switzerland (P.S.); and Kantonsspital Baselland, University of Basel, Basel, Switzerland (P.E.T.).

Acknowledgment: The authors thank all patients, physicians, and nurses associated with the SHCS.

Financial Support: By the framework of the SHCS, supported by grant 177499 from the Swiss National Science Foundation (SHCS project 842). Dr. Wandeler was supported by a professorship from the Swiss National Science Foundation (PP00P3 _176944). The data were gathered by the 5 Swiss university hospitals, 2 cantonal hospitals, 15 affiliated hospitals, and 36 private physicians (listed in www.shcs.ch/180-health-care-providers).

Disclosures: Disclosures can be viewed at www.acponline.org /authors/icmje/ConflictOfInterestForms.do?msNum=M20-4853.

Reproducible Research Statement: Study protocol and data set: Not available. *Statistical code:* Parts of the statistical code can be made available on reasonable request; contact Dr. Surial (e-mail, bernard.surial@insel.ch).

Corresponding Author: Bernard Surial, MD, Department of Infectious Diseases, Inselspital, Bern University Hospital, 3010 Bern, Switzerland; e-mail, bernard.surial@insel.ch.

Current author addresses and author contributions are available at Annals.org.

References

1. World Health Organization. Update of recommendations on first- and second-line antiretroviral regimens. July 2019. Accessed at www.who.int/hiv/pub/arv/arv-update-2019-policy/en on 19 June 2020.

2. Saag MS, Benson CA, Gandhi RT, et al. Antiretroviral drugs for treatment and prevention of HIV infection in adults: 2018 recommendations of the International Antiviral Society-USA panel. JAMA. 2018;320:379-396. [PMID: 30043070] doi:10.1001/jama.2018.8431

3. European AIDS Clinical Society. Guidelines for the treatment of HIV-positive adults. November 2019. Accessed at www.eacsociety .org/files/2019_guidelines-10.0_final.pdf on 18 March 2020.

4. Panel on Antiretroviral Guidelines for Adults and Adolescents. Guidelines for the Use of Antiretroviral Agents in Adults and Adolescents with HIV. Updated December 2019. Accessed at https://clinicalinfo.hiv.gov/sites/default/files/guidelines/documents /AdultandAdolescentGL.pdf on 4 June 2020.

5. Fux CA, Simcock M, Wolbers M, et al; Swiss HIV Cohort Study. Tenofovir use is associated with a reduction in calculated glomerular filtration rates in the Swiss HIV Cohort Study. Antivir Ther. 2007;12:1165-73. [PMID: 18240857]

6. Cooper RD, Wiebe N, Smith N, et al. Systematic review and meta-analysis: renal safety of tenofovir disoproxil fumarate in HIV-infected patients. Clin Infect Dis. 2010;51:496-505. [PMID: 20673002] doi:10.1086/655681

7. McComsey GA, Kitch D, Daar ES, et al. Bone mineral density and fractures in antiretroviral-naive persons randomized to receive abacavir-lamivudine or tenofovir disoproxil fumarate-emtricitabine along with efavirenz or atazanavir-ritonavir: Aids Clinical Trials Group A5224s, a substudy of ACTG A5202. J Infect Dis. 2011; 203:1791-801. [PMID: 21606537] doi:10.1093/infdis/jir188

8. Mills A, Crofoot G Jr, McDonald C, et al. Tenofovir alafenamide versus tenofovir disoproxil fumarate in the first protease inhibitorbased single-tablet regimen for initial HIV-1 therapy: a randomized phase 2 study. J Acquir Immune Defic Syndr. 2015;69:439-45. [PMID: 25867913] doi:10.1097/QAI.0000000000000618

9. Mills A, Arribas JR, Andrade-Villanueva J, et al; GS-US-292-0109 team. Switching from tenofovir disoproxil fumarate to tenofovir alafenamide in antiretroviral regimens for virologically suppressed adults with HIV-1 infection: a randomised, active-controlled, multicentre, open-label, phase 3, non-inferiority study. Lancet Infect Dis. 2016;16:43-52. [PMID: 26538525] doi:10.1016/S1473-3099(15) 00348-5

10. Surial B, Ledergerber B, Calmy A, et al; Swiss HIV Cohort Study. Changes in renal function after switching from TDF to TAF in HIV-infected individuals: a prospective cohort study. J Infect Dis. 2020;222:637-645. [PMID: 32189003] doi:10.1093/infdis/jiaa125

11. Gotham D, Hill A, Pozniak AL. Candidates for inclusion in a universal antiretroviral regimen: tenofovir alafenamide. Curr Opin HIV AIDS. 2017;12:324-333. [PMID: 28403027] doi:10.1097/COH .00000000000379

12. Venter WDF, Moorhouse M, Sokhela S, et al. Dolutegravir plus two different prodrugs of tenofovir to treat HIV. N Engl J Med. 2019;381:803-815. [PMID: 31339677] doi:10.1056/NEJMoa1902824 13. Sax PE, Erlandson KM, Lake JE, et al. Weight gain following initiation of antiretroviral therapy: risk factors in randomized comparative clinical trials. Clin Infect Dis. 2020;71:1379-1389. [PMID: 31606734] doi:10.1093/cid/ciz999

14. Yuh B, Tate J, Butt AA, et al. Weight change after antiretroviral therapy and mortality. Clin Infect Dis. 2015;60:1852-9. [PMID: 25761868] doi:10.1093/cid/civ192

 Schoeni-Affolter F, Ledergerber B, Rickenbach M, et al; Swiss HIV Cohort Study. Cohort profile: the Swiss HIV Cohort Study. Int J Epidemiol. 2010;39:1179-89. [PMID: 19948780] doi:10.1093/ije/dyp321
World Health Organization. Body mass index (BMI). Accessed at www.euro.who.int/en/health-topics/disease-prevention/nutrition /a-healthy-lifestyle/body-mass-index-bmi on 22 February 2021.

17. Verhaegen AA, Van Gaal LF. Drug-induced obesity and its metabolic consequences: a review with a focus on mechanisms and possible therapeutic options. J Endocrinol Invest. 2017;40:1165-1174. [PMID: 28660606] doi:10.1007/s40618-017-0719-6

18. The R Project for Statistical Computing. R Project. 2020. Accessed at www.R-project.org on 4 June 2020.

19. Santos JR, Saumoy M, Curran A, et al; Tenofovir/emtricitabine inflUence on LIPid metabolism (TULIP) Study Group. The lipid-lowering effect of tenofovir/emtricitabine: a randomized, crossover, double-blind, placebo-controlled trial. Clin Infect Dis. 2015;61:403-8. [PMID: 25870325] doi:10.1093/cid/civ296

20. Gomez M, Seybold U, Roider J, et al. A retrospective analysis of weight changes in HIV-positive patients switching from a tenofovir disoproxil fumarate (TDF)- to a tenofovir alafenamide fumarate (TAF)-containing treatment regimen in one German university hospital in 2015-2017. Infection. 2019;47:95-102. [PMID: 30269210] doi:10.1007/s15010-018-1227-0

21. Taramasso L, Berruti M, Briano F, et al. The switch from tenofovir disoproxil fumarate to tenofovir alafenamide determines weight gain in patients on rilpivirine-based regimen. AIDS. 2020;34:877-881. [PMID: 32271252] doi:10.1097/QAD.00000000002496

22. Hasse B, Iff M, Ledergerber B, et al; Swiss HIV Cohort Study. Obesity trends and body mass index changes after starting antiretroviral treatment: the Swiss HIV Cohort Study. Open Forum Infect Dis. 2014;1:ofu040. [PMID: 25734114] doi:10.1093/ofid/ofu040

23. McCormick CL, Francis AM, Iliffe K, et al. Increasing obesity in treated female HIV patients from Sub-Saharan Africa: potential causes and possible targets for intervention. Front Immunol. 2014; 5:507. [PMID: 25431572] doi:10.3389/fimmu.2014.00507

24. Taramasso L, Ricci E, Menzaghi B, et al; CISAI Study Group. Weight gain: a possible side effect of all antiretrovirals. Open

ORIGINAL RESEARCH

Forum Infect Dis. 2017;4:ofx239. [PMID: 29255735] doi:10.1093 /ofid/ofx239

25. Mayer KH, Molina JM, Thompson MA, et al. Emtricitabine and tenofovir alafenamide vs emtricitabine and tenofovir disoproxil fumarate for HIV pre-exposure prophylaxis (DISCOVER): primary results from a randomised, double-blind, multicentre, active-controlled, phase 3, non-inferiority trial. Lancet. 2020;396:239-254. [PMID: 32711800] doi:10.1016/S0140-6736(20)31065-5

26. Lacey A, Savinelli S, Barco EA, et al; UCD ID Cohort Study. Investigating the effect of antiretroviral switch to tenofovir alafenamide on lipid profiles in people living with HIV. AIDS. 2020;34:1161-1170. [PMID: 32310899] doi:10.1097/QAD .00000000002541

27. Sax PE, Wohl D, Yin MT, et al; GS-US-292-0104/0111 Study Team. Tenofovir alafenamide versus tenofovir disoproxil fumarate, coformulated with elvitegravir, cobicistat, and emtricitabine, for initial treatment of HIV-1 infection: two randomised, double-blind, phase 3, non-inferiority trials. Lancet. 2015;385:2606-15. [PMID: 25890673] doi:10.1016/S0140 -6736(15)60616-X

28. Randell PA, Jackson AG, Zhong L, et al. The effect of tenofovir disoproxil fumarate on whole-body insulin sensitivity, lipids and adipokines in healthy volunteers. Antivir Ther. 2010;15:227-33. [PMID: 20386078] doi:10.3851/IMP1518

29. Spinner CD, Schulz S, Bauer U, et al. Effects of antiretroviral combination therapies F/TAF, E/C/F/TAF and R/F/TAF on insulin resistance in healthy volunteers: the TAF-IR Study. Antivir Ther. 2018;23:629-632. [PMID: 30281025] doi:10.3851/IMP3271

30. Hill A, McCann K, Qavi A, et al. Risks of metabolic syndrome, diabetes, and cardiovascular disease in ADVANCE trial [Abstract]. Presented at Conference on Retroviruses and Opportunistic Infections, Boston, Massachusetts, 8-11 March 2020. Abstract no. 81.

Current Author Addresses: Drs. Surial, Mugglin, Furrer, Wandeler, and Rauch: Department of Infectious Diseases, Inselspital, Bern University Hospital, Freiburgstrasse 16p, 3010 Bern, Switzerland.

Dr. Calmy: Division of Infectious Diseases, Geneva University Hospital, Rue Gabrielle-Perret-Gentil 4, 1205 Geneva, Switzerland. Dr. Cavassini: Division of Infectious Diseases, University Hospital of Lausanne, Rue du Bugnon 46, 1011 Lausanne, Switzerland.

Drs. Günthard and Ledergerber: Division of Infectious Diseases and Hospital Epidemiology, University Hospital Zurich, Rämistrasse 100, 8091 Zürich, Switzerland.

Dr. Stöckle: Division of Infectious Diseases and Hospital Epidemiology, University Hospital Basel, Petersgraben 4, 4031 Basel, Switzerland.

Dr. Bernasconi: Division of Infectious Diseases, Regional Hospital of Lugano, Via Tesserete 46, 6900 Lugano, Switzerland.

Dr. Schmid: Division of Infectious Diseases, Cantonal Hospital of St. Gallen, Rorschacherstrasse 95, 9007 St. Gallen, Switzerland.

Dr. Tarr: Department of Medicine and Division of Infectious Diseases and Hospital Epidemiology, Kantonsspital Baselland, 4101 Bruderholz, Switzerland.

Author Contributions: Conception and design: B. Surial, C. Mugglin, G. Wandeler, A. Rauch.

Analysis and interpretation of the data: B. Surial, C. Mugglin, G. Wandeler, A. Rauch.

Drafting of the article: B. Surial, C. Mugglin, G. Wandeler, A. Rauch.

Critical revision of the article for important intellectual content: B. Surial, C. Mugglin, A. Calmy, M. Cavassini, H.F. Günthard, M. Stöckle, E. Bernasconi, P. Schmid, P.E. Tarr, H. Furrer, B. Ledergerber, G. Wandeler, A. Rauch.

Final approval of the article: B. Surial, C. Mugglin, A. Calmy, M. Cavassini, H.F. Günthard, M. Stöckle, E. Bernasconi, P. Schmid, P.E. Tarr, H. Furrer, B. Ledergerber, G. Wandeler, A. Rauch.

Provision of study materials or patients: A. Calmy, M. Cavassini,

H.F. Günthard, M. Stöckle, E. Bernasconi, P. Schmid, P.E. Tarr, H. Furrer.

Statistical expertise: B. Surial, B. Ledergerber.

Obtaining of funding: B. Surial, G. Wandeler, A. Rauch.

Administrative, technical, or logistic support: B. Surial, C. Mugglin, G. Wandeler, A. Rauch.

Collection and assembly of data: B. Surial, C. Mugglin, A. Calmy, M. Cavassini, H.F. Günthard, M. Stöckle, E. Bernasconi, P. Schmid, P.E. Tarr, H. Furrer, B. Ledergerber, G. Wandeler, A. Rauch.

Appendix: Members of the Swiss HIV Cohort Study

K. Aebi-Popp*, A. Anagnostopoulos*, M. Battegay*, E. Bernasconi†, J. Böni*, D.L. Braun*, H.C. Bucher*, A. Calmy[†], M. Cavassini[†], A. Ciuffi^{*}, G. Dollenmaier^{*}, M. Egger*, L. Elzi*, J. Fehr*, J. Fellay*, H. Furrer†, C.A. Fux*, H.F. Günthard[†] (president of the Swiss HIV Cohort Study), D. Haerry* (deputy of "Positive Council"), B. Hasse*, H.H. Hirsch*, M. Hoffmann*, I. Hösli*, M. Huber*, C.R. Kahlert* (chairman of the Mother and Child Substudy), L. Kaiser*, O. Keiser*, T. Klimkait*, R.D. Kouyos*, H. Kovari*, B. Ledergerber†, G. Martinetti*, B. Martinez de Tejada*, C. Marzolini*, K.J. Metzner*, N. Müller*, D. Nicca*, P. Paioni*, G. Pantaleo*, M. Perreau*, A. Rauch⁺ (chairman of the Scientific Board), C. Rudin^{*}, A.U. Scherrer* (head of Data Centre), P. Schmid†, R. Speck*, M. Stöckle† (chairman of the Clinical and Laboratory Committee), P. Tarr†, A. Trkola*, P. Vernazza*, G. Wandeler†, R. Weber*, and S. Yerly*.

* Members of the Swiss HIV Cohort Study who contributed to this work but did not author it.

† Members of the Swiss HIV Cohort Study who authored this work.





NRTI = nucleoside reverse transcriptase inhibitor; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Table 1. Missing Observations at Index Visit and Among All Follow-up Visits for Weight and Lipid Analyses

Variable	TDF, n (%)	TAF , n (%
Index visit assessments	891	3484
TDF vs. TAF	0	0
Sex	0	0
African vs. non-African origin	0	0
Weight	15 (1.7)	34 (1.0)
BMI	15 (1.7)	34 (1.0)
Age	0	0
Years since ART initiation	0	0
Third drug class switched to	12 (1.3)	16 (0.5)
Renal function (eGFR)	0	0
CD4 at index date	6 (0.7)	16 (0.5)
Smoker	1 (0.1)	0
Cholesterol	11 (1.2)	26 (0.7)
HDL cholesterol	17 (1.9)	75 (2.2)
LDL cholesterol	22 (2.5)	96 (2.8)
Triglycerides	16 (1.8)	47 (1.3)
Total-HDL cholesterol ratio	17 (1.9)	78 (2.2)
Follow-up visits	6246	25 009
Weight measurement	96 (1.5)	229 (0.9)
Physical activity	51 (0.8)	108 (0.4)
Weight-modifying drug	0	0
Smoking status	29 (0.5)	140 (0.6)
Total cholesterol	7 (0.1)	24 (0.1)
HDL cholesterol	37 (0.6)	106 (0.4)
LDL cholesterol	61 (1.0)	234 (0.9)
Triglycerides	24 (0.4)	125 (0.5)
Total cholesterol-HDL ratio	39 (0.6)	126 (0.5)
Use of a lipid-lowering drug	0	0

ART = antiretroviral therapy; BMI = body mass index; eGFR = estimated glomerular filtration rate; HDL = high-density lipoprotein; LDL = low-density lipoprotein; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.



Appendix Figure 2. Unadjusted distribution of weight over time.

Kernel density and box-and-whisker plots showing the distribution of weight measurements between TDF and TAF over time. TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Table 2. Estimates of the Multivariable Mixed-Effects Model for Weight Over Time

Fixed Effects	Estimate	SE	Statistic
Intercept	6.87	0.64	10.78
Time (years) 1	-6.5	0.66	-9.85
Time (years) 2	8.04	2.51	3.2
Time (years) 3	-32.24	9.61	-3.35
Time (years) 4	-55.67	19.68	-2.83
TAF vs. TDF	-16.79	64.99	-0.26
Index weight (per 10 kg)	8.96	0.06	148.07
Female vs. male	-0.42	0.1	-4.06
Non-African vs. African origin	-0.03	0.13	-0.26
Smoking vs. not smoking	-0.52	0.07	-7.63
Age per 10-y increase	0.09	0.05	1.91
Time since ART initiation (per year)	-0.01	0.01	-0.9
NNRTI vs. INSTI	-0.04	0.09	-0.51
PI vs. INSTI	-0.06	0.12	-0.49
CD4 at index date (per 100 cells/µL)	0.02	0.01	1.7
Physical activity: 1-4 times/mo vs. never	0.07	0.07	0.99
Physical activity: 1-2 times/wk vs. never	-0.24	0.06	-3.86
Physical activity: >2 times/wk vs. never	-0.39	0.06	-6.57
Use of weight-modifying drug	-0.04	0.12	-0.31
eGFR at index date (per 10-mL/min	0.04	0.02	1.51
change)			
Time (years) 1: TAF vs. TDF	19.52	64.59	0.3
Time (years) 2: TAF vs. TDF	6.45	37.8	0.17
Time (years) 3: TAF vs. TDF	44.27	124.63	0.36
Time (years) 4: TAF vs. TDF	48.42	36.12	1.34
Time (years) 1: Index weight (per 10 kg)	0.94	0.09	10.96
Time (years) 2: Index weight (per 10 kg)	-0.87	0.33	-2.64
Time (years) 3: Index weight (per 10 kg)	4.81	1.26	3.82
Time (years) 4: Index weight (per 10 kg)	8.06	2.58	3.13
TAF vs. TDF: Index weight (per 10 kg)	-2.34	8.41	-0.28
Time (years) 1: TAF vs. TDF: Index weight (per 10 kg)	2.16	8.36	0.26
Time (years) 2: TAF vs. TDF: Index weight (per 10 kg)	1.96	4.89	0.4
Time (years) 3: TAF vs. TDF: Index weight (per 10 kg)	2.58	16.14	0.16
Time (years) 4: TAF vs. TDF: Index weight (per 10 kg)	-4.51	4.71	-0.96
Random Effects		SD	
Patient ID nested in cohort center		2.18	
(random intercept)		20	

Years (random slope) 1.52 Residual 2.76 ART = antiretroviral therapy; eGFR = estimated glomerular filtration rate; INSTI = integrase strand transfer inhibitor; NNRTI = nonnucleo-

0.08

Cohort center (random intercept)

side reverse transcriptase inhibitor; PI = protease inhibitor; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Figure 3. Adjusted mean absolute weight over time.



Mean weight changes over time after switching to TAF compared with continuing TDF. Adjusted for sex; African origin; age; CD4 cell count; time since antiretroviral therapy initiation; index visit values of weight and estimated glomerular filtration rate; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. A total of 4291 individuals were included in the models. TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Table 3. Sensitivity Analyses of Patients With at Least 6 and 12 Months of Follow-up

Variable	Change in Weight After 18 mo of TDF (95% CI), <i>kg</i>	Change in Weight After 18 mo of TAF (95% CI), <i>kg</i>	Difference Between TAF and TDF (95% CI), <i>kg</i>	P Value for Group Difference
At least 6 mo of follow-up (n = 3821)	0.7 (0.4-0.9)	1.8 (1.6-2.0)	1.2 (0.8-1.5)	<0.001
At least 12 mo of follow-up (n = 3127)	0.6 (0.3-0.9)	1.9 (1.6-2.1)	1.2 (0.9-1.6)	<0.001

TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Figure 4. Adjusted mean weight over time, stratified by BMI category at the index visit.



Mean weight changes over time after switching to TAF compared with continuing TDF, stratified by BMI categories at the index visit. Adjusted for sex; African origin; age; CD4 cell count; time since antiretroviral therapy initiation; index visit values of weight and estimated glomerular filtration rate; third drug used after the index visit; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. A total of 4291 individuals were included in the models. BMI = body mass index; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Figure 5. Changes in weight over time, analysis restricted to patients who had a suppressed HIV viral load 1 y before baseline and through follow-up (includes 758 who continued TDF and 2930 who switched to TAF).



According to a multivariable model, adjusted for sex; African origin; baseline body mass index, age, and CD4 cell count; time since antiretroviral therapy initiation; third drug used after baseline; and time-updated use of weight-modifying drugs, smoking status, and physical activity. The model includes random intercepts for each patient nested within cohort site and a random slope for time on the individual level. TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Appendix Table 4. Sensitivity Analysis of Patients Who Had Only TDF Replaced With TAF, Without Further Changes in Antiretroviral Therapy*

Variable	Change in Weight After 18 mo of TDF (95% Cl), <i>kg</i>	Change in Weight After 18 mo of TAF (95% CI), <i>kg</i>	Difference Between TAF and TDF (95% CI), <i>kg</i>	<i>P</i> Value
Overall	0.6 (0.3 to 0.9)	1.6 (1.3 to 1.9)	1.0 (0.6 to 1.4)	< 0.001
INSTI	0.0 (-0.6 to 0.6)	1.5 (1.1 to 1.9)	1.5 (0.7 to 2.2)	< 0.001
NNRTI	0.8 (0.5 to 1.1)	1.9 (1.4 to 2.5)	1.1 (0.5 to 1.8)	0.001
PI	0.8 (-0.1 to 1.8)	1.9 (1.2 to 2.5)	1.0 (-0.1 to 2.1)	0.081

INSTI = integrase strand transfer inhibitor; NNRTI = nonnucleoside reverse transcriptase inhibitor; PI = protease inhibitor; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

* Includes 885 who continued TDF and 2022 who switched to TAF.

Appendix Table 5. Adjusted Mean Differences of Lipid Levels After 18 Months Compared With the Index Visit After Switching From TDF to TAF or Continuing TDF*

Variable	Mean Difference in Those Who Continued TDF (95% CI)	Mean Difference in Those Who Switched to TAF (95% CI)
Total cholesterol mmol/L mg/dL	-0.09 (-0.15 to -0.03) -3.5 (-5.6 to -1.3)	0.24 (0.19 to 0.30) 9.5 (7.4 to 11.5)
HDL cholesterol mmol/L mg/dL	-0.01 (-0.03 to 0.01) -0.3 (-1.0 to 0.3)	0.05 (0.03 to 0.06) 1.9 (1.2 to 2.5)
LDL cholesterol mmol/L mg/dL	-0.07 (-0.12 to -0.02) -2.7 (-4.7 to -0.8)	0.12 (0.07 to 0.17) 4.7 (2.8 to 6.6)
Triglycerides mmol/L mg/dL	-0.05 (-0.12 to 0.03) -4.0 (-10.6 to 2.5)	0.18 (0.11 to 0.25) 16.0 (9.8 to 22.3)
Total cholesterol-HDL ratio	-0.07 (-0.14 to 0.01)	0.05 (-0.02 to 0.12)

HDL = high-density lipoprotein; LDL = low-density lipoprotein; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

* Adjusted for age; sex; African origin; each lipid level at the index visit; and time-varying physical activity, weight, and use of lipid-lowering medication. The models include random intercepts for each patient nested within cohort site and a random slope for time on the individual level.

Appendix Table 6. Incidence Rate Ratios of New-Onset
Diabetes Among Individuals Without Preestablished
Diabetes at the Index Visit*

IRR (95% CI)	P Value
1.3 (0.7-2.8)	0.49
1.5 (1.2-1.9)	< 0.001
0.4 (0.2-0.9)	0.037
0.6 (0.3-1.6)	0.31
1.9 (1.5–2.2)	< 0.001
	IRR (95% CI) 1.3 (0.7-2.8) 1.5 (1.2-1.9) 0.4 (0.2-0.9) 0.6 (0.3-1.6) 1.9 (1.5-2.2)

BMI = body mass index; IRR = incidence rate ratio; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

n = 4150, adjusted for all categories in the table.

Appendix Figure 6. Numbers of individuals (percentages) with new onset of diabetes among persons without diabetes at the index visit stratified by index visit BMI category (n = 4150).



BMI = body mass index; TAF = tenofovir alafenamide; TDF = tenofovir disoproxil fumarate.

Annals of Internal Medicine